
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
Sat Aug 12 18:23:14 GMT 1995
Year: 95  Doy: 224
Observer: koon       console      Aug 12 18:19
Content-Length: 819
X-Lines: 29
Status: RO

WEATHER COMMENT: Sat Aug 12 18:23:19 GMT 1995
Passing fog, wind=10 mph from the SE, temp=43 F.

LOW-L COMMENT: Sat Aug 12 18:30:37 GMT 1995
Removed L00293 and installed L00295 in drive #1

Sat Aug 12 18:34:23 GMT 1995: Patrol Start

WEATHER COMMENT: Sat Aug 12 18:50:47 GMT 1995
Fog has lifted so I am running observations, but still have passing low clouds

Sat Aug 12 19:01:56 GMT 1995: Calibration
Sat Aug 12 19:55:58 GMT 1995: Patrol End

WEATHER COMMENT: Sat Aug 12 19:56:03 GMT 1995
Starting to get foggy again, paused instruments and closed dome.

WEATHER COMMENT: Sun Aug 13 01:08:11 GMT 1995
Weather didn’t fully clear, no data on tapes (except Low-l), recycling them.

COMMENT: Sun Aug 13 01:09:29 GMT 1995
TAPES:
******
LOW-L: L00294 in drive #0, no crashes today.

Sun Aug 13 01:10:05 GMT 1995: Filemark


